Author's response to reviews

Title: Implementation of preventive strength training in residential geriatric care: a multi-centre study protocol with one year of interventions on multiple levels

Authors:

Michael Brach (michael.brach@uni-muenster.de)
Frank Nieder (f.nieder@dshs-koeln.de)
Ulrike Nieder (u.nieder@dshs-koeln.de)
Heinz Mechling (mechling@dshs-koeln.de)

Version: 6 Date: 18 November 2009

Author's response to reviews:

Dear Editor.

A change in the Authors' contributions section was required.

I expanded on the reference in the authors contributions sections so that all the details are available. The ref. was deleted in the reference list, so that old #67 changed to #67.

No further changes were made.

Kind regards,

Michael Brach